Author’s response to reviews

Title: Haemodynamics and oxygenation improvement induced by High Frequency Percussive Ventilation in a hypoxic cardiac surgery patient.

Authors:

alessandro forti (alefortidoc@me.com)
valeria salandin (vsalandin@ulss.tv.it)
paolo zanatta (pzanatta@mac.com)
bruno persi (bpersi@ulss.tv.it)
carlo sorbara (csorbara@ulss.tv.it)

Version: 2 Date: 15 February 2010

Author’s response to reviews:

Dear Editorial Office,

we thank you very much for identifying this missing information in the manuscript. This is an excellent suggestion and we have restructured the abstract in the introduction and in the conclusion.

We would like to thank the Editor for identifying this important limitation in our manuscript and it is with great pleasure that we resubmit it for consideration of peer review in journal of medical case report.

Alessandro Forti MD